Author’s response to reviews

Title: The Effect of Mild Agonist Stimulation on the Platelet Reactivity in Patients with Type 2 Diabetes mellitus

Authors:
Razie Mahmoodian (r.mahmoodiyan@gmail.com)
Morteza salimian (salimian.mrtz@gmail.com)
Mohsen Hamidpour (mohsenhp@sbmu.ac.ir)
Ali Akbar Khadem-Mabudi (Maboudi_3@yahoo.com)
Ahmad Gharehbaghian (gharehbaghian@sbmu.ac.ir)

Version: 2 Date: 31 May 2019

Author’s response to reviews:

Dear Esther Fagelson,

Thank you for your email. I would like to appreciate for your useful and effective comments.

1. As you recommended, the manuscript has been revised and edited by professional language editors.
2. It is necessary to say that in the Author’s contributions section of manuscript, AH had been typed instead of AG wrongly. So it has been corrected in final version and refers to Ahmad Gharehbaghian.
3. The final and clean version of manuscript is uploaded.

Sincerely yours